



May 27, 2015

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 19339-review.doc).

**Title: Are stapler line reinforcement materials necessary in sleeve gastrectomy?**

**Author:** İbrahim Sakçak

**Name of Journal:** *World Journal of Surgical Procedures*

**ESPS Manuscript NO:** 19339

The manuscript has been improved according to the suggestions of reviewers:

1 Format has been updated

2 Revision has been made according to the suggestions of the reviewer

(1) **revised my manuscript according to the reviewers' and editor's suggestions**

(2) **Upload Reviewers response letter**

(3) **Upload Copyright Assignment form**

(4) **Upload Cor tip audio file**

3 References and typesetting were corrected

Thank you again for publishing our manuscript in the *World Journal of Surgical Procedures*.

Sincerely yours,

A handwritten signature in black ink, appearing to be 'İbrahim Sakçak'.

Assos. Prof.Dr.İbrahim Sakçak, MD

Department of General Surgery

Hospital of Medicalpark

Ankara, TURKEY

Fax: +90-312-6668169

E-mail: [ibrahimsakcak66@gmail.com](mailto:ibrahimsakcak66@gmail.com)